Title: Effect of yogurt containing polydextrose, Lactobacillus acidophilus NCFM and Bifidobacterium lactis HN019: a randomized, double-blind, controlled study in chronic constipation

Authors:

Daniela O Magro (danimagro@terra.com.br)
Lais M Oliveira (laismariananao@hotmail.com)
Isabela Bernasconi (isabelabernasconi@hotmail.com)
Marilia Souza Ruela (mariliaruela04@gmail.com)
Laura Credidio (laurabio@gmail.com)
Irene K Barcelos (ibarcelos@fcn.unicamp.br)
Raquel F Leal (raquelleal@mpc.com.br)
Maria de Lourdes S Ayrizono (luayrizono@terra.com.br)
João J Fagundes (jj.fagundes@terra.com.br)
Leandro B Teixiera (leandro.teixeira@danisco.com)
Arthur C Ouwehand (Arthur.Ouwehand@dupont.com)
Claudio S Coy (claudiocoy@gmail.com)

Version: 2
Date: 22 April 2014

Author's response to reviews: see over
Cover letter

Nutrition Journal

Nehme Gabriel

Dear Dr.

We are submitting to Nutrition Journal a manuscript regarding the effect of yogurt containing probiotics in chronic constipation. All the authors have approved the submitted version of the manuscript and also declare that the manuscript, including related data, figures and tables has not been previously published and that the manuscript is not under consideration elsewhere.

Please note that we have made some changes to the formatting of our layout manuscript.

Sincerely,

Claudio S.R. Coy